Author's response to reviews

Title: Kaliziri extract upregulates Tyrosinase, TRP-1, TRP-2 and MITF expression in Murine B16 Melanoma cells

Authors:

Adila Tuerxuntayi (adila1120@hotmail.com)
Yong-qiang Liu (johney1983@gmail.com)
Ablajan Tulake (ablj1228@sina.com)
Maidina kabas (maidina27@hotmail.com)
Eblimit Aiden (aabulimi@bcm.edu)
Haji Akber Aisa (haji@ms.xjb.ac.cn)

Version: 6 Date: 9 April 2014

Author's response to reviews: see over
Dear Editor,

We are submitting our manuscript entitled “Kaliziri extract upregulates Tyrosinase, TRP-1, TRP-2 and MITF expression in Murine B16 Melanoma Cells” for your review. Kaliziri extract (KZE) is a traditional Uyghur medicine, used by Traditional hospitals for the treatment of vitiligo for more than 30 years in China; clinical application has showed that this medicine has obvious therapeutic effect. However, its phytochemical analysis and mechanism of its effectiveness had not been examined. In our study, the positive effect of KZE on melanogenesis in B16 cell lines was investigated to clarify the underlying molecular mechanism of melanin synthesis. The result indicates that KZE induced melanogenesis by increasing expression of tyrosinase, TRP-1 and TRP-2 via MITF in B16 cell. We believe that this work will be of general interest to BMC Complementary and Alternative Medicine’s readership as it provides important insights into the mechanism of action of traditional Uyghur medicines in the treatment of Vitiligo.

It is a great pleasure for me to received Reviewer comments and we revise the article conscientiously.

Author’s response to reviews: see over
Author's response to reviews

Title: Kaliziri extract upregulates Tyrosinase, TRP-1, TRP-2 and MITF expression in Murine B16 Melanoma cells

Authors:

Adila Tuerxuntayi (adila1120@hotmail.com)

Yong-qiang Liu (johney1983@gmail.com)

Ablajan Tulake (ablj1228@sina.com)

Maidina kabas (maidina27@hotmail.com)

Eblimit Aiden (aabulimi@bcm.edu)

Haji Akber Aisa (haji@ms.xjb.ac.cn)

Reviewer's report

Version: 4  Date: 9 March 2014

Reviewer: Sukanya Dej-adisai

Reviewer's report:

Minor Essential Revisions

Level of interest: An article of importance in its field

Quality of written English: Acceptable

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests: I declare that I have no competing interests

Reviewer's report:

1. Correction the writing style of scientific plant names such as Vernonia anthelmintica (L.) Willd. Because Willd. is the short name of the author. ### do not forget the full stop after
short name of the author (page 2, 6).

We correction the writing style of scientific plant names in page2, 6.

2. Reference 3: the name of pharmacopoeia should be bold same as the other references.


3. Correction the space of typing in all of the references i.e. after full stop should have one space, after journal’s name should have one space before the year, etc.

We correction the space of typing in all of the references.

4. Reference 14: collection the typing style of the journal.


5. Please provide details in your Methods section on who undertook the formal identification of the plant material used in your study.

We give the details in our Methods about identification information: Kaliziri [*Vernonia anthelmintica* (L.) Willd.] seeds (identify by Prof. Guan-mian Sheng from Xinjiang Institute of Ecology and Geography, Chinese Academy of Sciences)

6. However, many mistakes were showed in the references and text, so please correct them once more.

We use professional language editing service to improve the quality of the manuscript.

I really appreciate your precious advice. I will do my best to improve the level of my article.

We especially declare here that the abovementioned paper is not simultaneously submitted for publication elsewhere.

Authors declared that they have no competing interests.
Thank you for consideration!

We would also like to suggest the following reviewers based on their expertise in phytochemical research and Traditional Uyghur Medicine.

Yours sincerely,

Haji Akbar Aisa  Tel.: +86 18999212221

The Key Laboratory of Plant Resources and Chemistry of Arid Zone, Xinjiang Technical Institute of Physics and Chemistry, Chinese Academy of Sciences, Urumqi 830011

China State Key Laboratory Basis of Xinjiang Indigenous Medicinal Plants Resource Utilization, Urumqi 830011, China